Author’s response to reviews

Title: Adherence of Iron and folic acid supplementation and determinants among pregnant women in Ethiopia: A systematic review and Meta-analysis

Authors:

Melaku desta (melakd2015@gmail.com;melakd2018@gmail.com)
Bekalu Kassie (bekalukassiedmu@gmail.com)
Habtamu Chanie (habitclinton@gmail.com)
Henok Mulugeta (mulugetahenok68@gmail.com)
Tadesse Yirga (tadesseyirga680@gmail.com)
Habtamu Temesgen (habtamutem@gmail.com)
Cheru Leshargie (chertesema@gmail.com)
Yoseph Merkeb (ymerkeb11@gmail.com)

Version: 3 Date: 02 Nov 2019

Author’s response to reviews:

Dear Editors of Reproductive Health

We would like to extend our deepest appreciation for devoting your time to review our manuscript entitled “adherence of iron and folic acid supplementation among pregnant women in Ethiopia: A systematic review and meta-analysis”. Iron and folic acid deficiency anemia is a global public problem that causes maternal morbidity, which can be prevented through improving the adherence of iron-folic acid supplementation among pregnant women. Even though, the adherence is inconsistent across the country. Therefore, this systematic review and meta-analysis estimates the pooled adherence of Iron and folic acid supplementation and determinants among pregnant women in Ethiopia.

Dear reviewer, there has been a minor editing or revision of this manuscript (Abstract, introduction, methods, results, discussion and conclusions). Hence, “Based on these reports, and my own assessment as Editor, I am pleased to inform you that it is potentially acceptable for publication in Reproductive Health, once you have carried out some essential revisions suggested by our reviewers”. There is no listed comments by reviewers other than yours minor editing. The language has been extensively examined to correct grammatical and spelling inconsistencies again and the whole structure of the manuscript has been revised. We hope now the manuscript is clear and more acceptable than its previous version. We have tried to present
the paper in proper manner according to your comment what to supposed to do so. For this, here we have given our responses to each of the concerns you raised, highlighted by red color. Again, we would like to remind our strongest gratitude for your effort for the improvement of this manuscript and all the points were addressed in the point by point response.

Regards